Author's response to reviews

Title: Evaluation of health promotion in schools: a realistic evaluation approach using mixed methods

Authors:

Jeanine Pommier (Jeanine.Pommier@ehesp.fr)
Marie-Renée Guevel (Marie-Renee.Guevel@ehesp.fr)
Didier Jourdan (Didier.Jourdan@univ-bpclermont.fr)

Version: 3 Date: 15 January 2010

Author's response to reviews:

Dear Editor,

We did the following formatting changes requested:

- Table 2: We are unable to display the arrows in table 2. You may wish to upload it as a figure file and cite it as such within the manuscript text. please remember to renumber any subsequent tables or figures.

We upload table to as figure 3 and renumber the other tables and figures.

- Authors, affiliations, email addresses: Please list these separately on the title page. Authors should be linked to their affiliations with the use of superscript numbers and email addresses should be preceded by the initials of the authors.

Author list: please amend the capitalisation of your author list. The preferred style is "First name middle Initial Last name" (e.g. Joe F Bloggs)

We made the changes requested.

- Keywords: Please remove these.

We removed the keywords

- Tables: the table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table

We checked all the tables, the titles now appear above the tables and are only one sentence long.

- Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements
We re-layout table 2 (former table 3) as requested.

- Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

We checked the figures and provided them as a ppt format.

- Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.

We checked our manuscript for typographical errors.

We hope we have answered your request. Let us know if you have any other requests.

Thank you for your time and consideration.

Yours sincerely,

The authors

Jeanine Pommier, Marie-Renée Guevel, Didier Jourdan